Author’s response to reviews

Title: Why Acute Ischemic Stroke Patients in the United States Use or Do Not Use Emergency Medical Services Transport? Findings of an Inpatient Survey

Authors:

Sudha Xirasagar (sxirasagar@sc.edu)
Meng-han Tsai (metsai@csumb.edu)
Khosrow Heidari (heidarik@dhec.sc.gov)
James Hardin (jhardin@mailbox.sc.edu)
Yuqi Wu (yuqiw@email.sc.edu)
Robert Wronski (wronskra@dhec.sc.gov)
Dana Hurley (dana@huthglobal.com)
Edward Jauch (edward.jauch@HCAHealthcare.com)
Souvik Sen (souvik.sen@uscmed.sc.edu)

Version: 2 Date: 31 Oct 2019

Author’s response to reviews:

Item-wise responses to comments requests in Editor’s communication titled: Your submission to BMC Health Services Research - BHSR-D-19-01860R1

1. Please move the 'Abbreviations' to come after the 'Conclusion' section. – Completed

2. Please move the 'Declarations' to come after the 'Abbreviations' section. – Completed

3. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case. – Completed. Approvals were obtained from all authors after circulating the final manuscript with changes indicated above.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Please ensure all figures,
tables and additional files are cited in the text, as they may be removed by the production department. Please delete any additional files not intended for publication.

Response: All figures, tables appendices were checked for accuracy and cross-referenced correctly in the text.